Author’s response to reviews

Title: Mucopolysaccharidosis Type VI: Case Report with First Neonatal Presentation with Ascites Fetalis and Rapidly Progressive Cardiac Manifestation

Authors:

Rachel Honjo (rachel.honjo@hc.fm.usp.br)
Evelyn Nuñez Vaca (cristinanunezvaca@gmail.com)
Gabriela Leal (gnleal@gmail.com)
Deipara Abellan (deipara.abellan@hc.fm.usp.br)
Nana Ikari (nana.miura@hc.fm.usp.br)
Marcelo Jatene (marcelo.jatene@hc.fm.usp.br)
Ana Martins (ni.martins10@gmail.com)
Chong Kim (chong.kim@hc.fm.usp.br)

Version: 3 Date: 31 Jan 2020

Author’s response to reviews:

São Paulo, Brazil

January 31st, 2020

Dear Editorial Board of BMC Medical Genetics,

I would like to thank the opportunity to review and re-submit our case report.

Please find below the answers to the Editor comments of the manuscript “Mucopolysaccharidosis Type VI: Presentation with Ascites Fetalis and Rapidly Progressive Cardiac Manifestation”.

FORMATTING CHANGES:

1. We notice an identifying image is provided in Figure 1. Please crop this image to display only necessary portions that are relevant to the manuscript to ensure anonymity.
ANSWER: I checked the last version of Figure 1 and the patient’s eyes were already cropped. Thus I did not understand this last request. Email to the Editor in 12/11/2019 explaining that the Figure was already cropped and email received on Jan 6th asked to re-submit the revision in the online system.

New Figure 1 (with more parts cropped) was uploaded.

Sincerely yours,

Rachel Sayuri Honjo on behalf of the authors

Corresponding author:
Rachel Sayuri Honjo, M.D., Ph.D.
Unidade de Genética do Instituto da Criança – HCFMUSP
Av. Dr. Enéas Carvalho de Aguiar, 647
São Paulo – SP – Brazil
CEP 05403-000
Email: rachel.honjo@hc.fm.usp.br
Telephone: +55 (11) 2661-8671